Author's response to reviews

Title: Body mass index, physical activity, and dietary behaviors among members of an urban community fitness center: a questionnaire survey

Authors:

Kimberly A Kaphingst (kkaphing@mail.nih.gov)
Gary G Bennett (Gary_Bennett@dfci.harvard.edu)
Glorian Sorensen (Glorian_Sorensen@dfci.harvard.edu)
Karen M Kaphingst (kaphingst@epi.umn.edu)
Amy E O'Neil (Amy.ONeil@bcbsma.com)
Kyle McInnis (Kyle.Mcinnis@umb.edu)

Version: 4 Date: 16 July 2007

Author's response to reviews:

July 16, 2007

Please find attached a copy of our revised manuscript (MS: 8728857781292900) entitled "Body mass index, physical activity, and dietary behaviors among members of an urban community fitness center: A questionnaire survey."

As requested, we have:

- inserted the header "Tables" ahead of the tables section
- made a final check of the ms

If you have any questions or require additional information, please do not hesitate to contact me. Thank you for your consideration.

Best regards,

Kimberly A. Kaphingst, ScD
Investigator, Social and Behavioral Research Branch
National Human Genome Research Institute
Building 31, Room B1B37E
31 Center Drive, MSC 2073
Bethesda, MD 20892 USA
Ph: 301-451-4010
Fax: 301-480-3108
E-mail: kkaphing@mail.nih.gov